Author's response to reviews

Title: Effectiveness of a nutrition education package in improving feeding practices, dietary adequacy and growth of infants and young children in rural Tanzania: rationale, design and methods of a cluster randomised trial

Authors:

Kissa B.M. Kulwa (kissakulwa@yahoo.com)
Roosmarijn Verstraeten (rverstraeten@itg.be)
Kimberley P. Bouckaert (kimberley.bouckaert@ugent.be)
Peter S. Mamiro (petermamiro@yahoo.com)
Patrick W. Kolsteren (pkolsteren@itg.be)
Carl Lachat (carl.lachat@ugent.be)

Version: 2
Date: 6 October 2014

Author's response to reviews:

Ethics and funding approval documentation have been forwarded to BMC Series Editorial office.